ENHANCEMENT OF KAOLINITE USING SUPERABSORBENT POLYMER (SAP) AS EARTH GROUNDING MATERIAL

TEH CHEE YEE

B. ENG (HONS.) CIVIL ENGINEERING

UNIVERSITI MALAYSIA PAHANG

UNIVERSITI MALAYSIA PAHANG

DECLARATION OF THESIS AND COPYRIGHT							
Author's Full Name	: TEH CHEE YEE						
Date of Birth	: 24 th J	: 24 th JUNE 1993					
Title	: ENHA SUPEF GROU	: ENHANCEMENT OF KAOLINITE USING SUPERABSORBENT POLYMER (SAP) AS EARTH GROUNDING MATERIAL					
Academic Session	: 2016/	2017					
I declare that this thesi	s is class	ified as:					
CONFIDENTI	AL	(Contains confidential information under the Official					
□ RESTRICTED		(Contains restricted information as specified by the					
☑ OPEN ACCES	PEN ACCESSDescription organization where research was done)*I agree that my thesis to be published as online open acceleration (Full Text)						
I acknowledge that Un	iversiti N	Aalaysia Pahang reserves the following rights:					
 The Thesis is the Pr The Library of Univ the purpose of resea The Library has the 	coperty of versiti Ma arch only right to	f Universiti Malaysia Pahang alaysia Pahang has the right to make copies of the thesis for make copies of the thesis for academic exchange.					
Certified by:							
(Student's Sign	(Student's Signature) (Supervisor's Signature)						
New IC/Passport NumberName of SupervisorDate:Date:							

NOTE : * If the thesis is CONFIDENTIAL or RESTRICTED, please attach a thesis declaration letter.



SUPERVISOR'S DECLARATION

I hereby declare that I have checked this thesis and in my opinion, this thesis is adequate in terms of scope and quality for the award of the degree of Bachelor of Civil Engineering and Earth Resources (Hons.).

(Supervisor's Signature) Full Name : DR. HJ MOHD YUHYI BIN HJ MOHD TADZA Position : SENIOR LECTERUR Date : 19th JUNE 2017



STUDENT'S DECLARATION

I hereby declare that the work in this thesis is based on my original work except for quotations and citations which have been duly acknowledged. I also declare that it has not been previously or concurrently submitted for any other degree at Universiti Malaysia Pahang or any other institutions.

(Student's Signature) Full Name : TEH CHEE YEE ID Number : AA13145 Date : 19th JUNE 2017

ENHANCEMENT OF KAOLINITE USING SUPERABSORBENT POLYMER (SAP) AS EARTH GROUNDING MATERIAL

TEH CHEE YEE

Thesis submitted in fulfillment of the requirements for the award of the Bachelor Degree in Civil Engineering

Faculty of Civil Engineering and Earth Resources

UNIVERSITI MALAYSIA PAHANG

JUNE 2017

ACKNOWLEDGEMENTS

First and foremost, I would like to take this opportunity to thank University Malaysia Pahang for providing me the good condition of facilities and equipment to complete this study.

Following I would like to express my deepest gratitude to my supervisor, Dr. Hj Mohd Yuhyi Bin Hj Mohd Tadza who has been very supportive in guiding and providing me informative discussion and recommendation throughout the study. Without his patient guidance and dedicated involvement in every step throughout the process, this paper would have never been accomplished. I would like to thank you very much for your support and understanding.

In addition, I would like to express my gratitude to my panels, Dr. Sulistyo Arintono and Dr. Youventharan A/L Duraisamy to their valuable suggestions and comments on my work to allow me to improve my research outcome and meet the objectives of this study.

Apart from that, I would like to thank all the lecturers whom have taught me in every semester. They have indeed helped me to reinforce my basic knowledge and theories in this field.

A special thanks to my friend, Mohamad Hazman Bin Mohamad Yusof and Noor Balqish Binti Mohd Din for they are always willing to share their precious knowledge and resources with me in completing this research work

Finally, I am also grateful to my family members for their love and encouragement all the way to the completion of my task in this programme.

ABSTRAK

Sistem pembumian berfungsi sebagai perlindungan kilat dalam kilang perindustrian dan loji kuasa. Salah satu parameter utama untuk mengenal pasti prestasi sistem pembumian adalah kerintangan tanah tempatan. Kaolinit mempunyai keplastikan dan pengekalan air yang lebih rendah daripada air bentonit. Oleh itu, Superabsorbent Polymer (SAP), bahan dengan penyerapan cecair yang tinggi dan keupayaan pengekalan cecair ditambah dalam kaolinit untuk meningkatkan prestasi kaolinit dalam keplastikan dan kerintangan. Objektif utama kajian ini adalah untuk menentukan peningkatan dalam ciri keplastikan kaolinit dan mewujudkan ciri serapan air lengkung dengan penambahan SAP dalam pelbagai peratusan ke kaolinit. Kajian ini juga membandingkan kerintangan kaolinit dalam pelbagai peratusan SAP. Tiga jenis sampel kaolinit yang digunakan dalam kajian ini ialah S300, Speswhite dan FMC dengan campuran 10% SAP untuk menentukan ciri-ciri keplastikan. Had keplastikan untuk S300, Speswhite dan FMC dengan tambahan 10% SAP adalah 150.38%, 141.45% dan 275.30%. Indeks keplastikan untuk S300, Speswhite dan FMC dengan 10% daripada SAP adalah 13.10%, 19.40% dan 30.60%. Kerintangan S300, Speswhite dan FMC dengan tambahan 10% SAP adalah 130 Ω.m, 17 Ω.m dan 21 Ω.m. kerintangan ini adalah lebih tinggi daripada kerintangan bentonit dan Marconite, iaitu 6.2 Ω .m dan 0.38 Ω .m.

ABSTRACT

Earth grounding system serves as lightning-protection in industrial and power plants. One of the major parameters to determine the performance of the grounding system is the resistivity of local soil. Kaolinite has lower plasticity and lower retention of water as compared to bentonite. Thus, Superabsorbent Polymer (SAP), a material with high absorption and retention capacity of fluids is added to kaolinite to improve the performance of kaolinite in plasticity and resistivity. The main objectives of the study were to determine the improvement in the plasticity characteristic of kaolinite and establish soil-water characteristics curve (SWCC) by addition of SAP at varying percentage. This study also compare the resistivity of kaolinite at varying percentage of SAP. Three kaolinite samples are used in this study which are S300, Speswhite and FMC with the mixture 10% of SAP to determine the plasticity characteristic. The plastic limit for S300, Speswhite and FMC with addition of 10% SAP are 150.38%, 141.45% and 275.30%. The plasticity index for S300, Speswhite and FMC with 10% of SAP are 130 Ω .m, 17 Ω .m and 21 Ω .m respectively. These resistivity are higher than resistivity of bentonite and Marconite, which are 6.2 Ω .m and 0.38 Ω .m.

TABLE OF CONTENT

DEC	CLARATION	
TIT	LE PAGE	
ACK	KNOWLEDGEMENTS	ii
ABS	TRAK	iii
ABS	TRACT	iv
TAB	BLE OF CONTENT	V
LIST	Г OF TABLES	viii
LIST	Γ OF FIGURES	ix
LIST	Г OF SYMBOLS	X
LIST	Γ OF ABBREVIATIONS	xi
CHA	APTER 1 INTRODUCTION	1
1.1	Background	1
1.2	Problem Statement	3
1.3	Research Objectives	3
1.4	Scope of Study	4
1.5	Significance of research	4
1.6	Thesis overview	4
CHA	APTER 2 LITERATURE REVIEW	6
2.1	Introduction	6
2.2	Earth Grounding System	6
	2.2.1 Importance of Earth Grounding System	6

	2.2.2	Earth Grounding Resistance	7
	2.2.3	Grounding Enhancement Material	9
2.3	Clay		10
	2.3.1	Bentonite	10
	2.3.2	Kaolinite	12
	2.3.3	Marconite	13
2.4	Supera	absorbent Polymer (SAP)	14
2.5	Soil W	Vater Characteristic Curve (SWCC)	15
CHA	PTER 3	METHODOLOGY	18
3.1	Introd	uction	18
3.2	Select	ion of Material	18
	3.2.1	Clay	18
	3.2.2	Superabsorbent Polymer (SAP)	18
3.3	Soil S	ample Preparation	18
3.4	Physic	cal Properties of Kaolinite	19
	3.4.1	Liquid Limit, LL	19
	3.4.2	Plastic Limit, PL	20
	3.4.3	Shrinkage Limit	20
	3.4.4	Specific Gravity	21
	3.4.5	Swelling Index	21
	3.4.6	Specific Surface Area	22
	3.4.7	Loss of Ignition	23
3.5	Chem	ical Property	23
	3.5.1	Cation-Exchange Capacity (CEC)	23
3.6	Soil-V	Vater Characteristic Curve (SWCC)	23

CHAI	PTER 4	RESULTS AND DISCUSSION	26
4.1	Introd	uction	26
4.2	Physic	cal properties	26
	4.2.1	Liquid Limit	26
	4.2.2	Plastic Limit	27
	4.2.3	Shrinkage Limit	28
	4.2.4	Specific Gravity	30
	4.2.5	Swelling Index	30
	4.2.6	BET Surface Area	31
	4.2.7	Loss on Ignition (LOI)	31
	4.2.8	Particle size distribution	32
4.3	Chem	ical Property	32
	4.3.1	Cation-Exchange Capacity (CEC)	32
4.4	Soil W	Vater Characteristic Curve	34
4.5	Soil R	esistivity	36
CHAI	PTER 5	CONCLUSION	37
5.1	Concl	usion	37
5.2	Recon	nmendations	38
REFE	RENC	ES	39

24

LIST OF TABLES

Soil resistivity and earthing resistance required for different type of	
soil	8
Effect of moisture content on soil resistivity	8
Water absorption of bentonite and kaolinite from different places	11
Properties of kaolin (kaolinite) and smectite (bentonite)	13
Properties of S300, Speswhite, FMC, Bentonite and Marconite with	
varying percentage of SAP	33
Soil resistivity of various soil samples	36
	Soil resistivity and earthing resistance required for different type of soil Effect of moisture content on soil resistivity Water absorption of bentonite and kaolinite from different places Properties of kaolin (kaolinite) and smectite (bentonite) Properties of S300, Speswhite, FMC, Bentonite and Marconite with varying percentage of SAP Soil resistivity of various soil samples

LIST OF FIGURES

Diagrammatic sketch of the structure of kaolinite	12
Marconite installation in grounding system	14
Schematic diagram of a dew-point Water PotentiaMeter, WP4	16
Superabsorbent polymer before oven drying	19
Superabsorbent polymer after oven drying at 60° c	19
Shrinkage limit test for S300, FMC and Speswhite with addition of 10% SAP $$	20
Swelling index for S300, Speswhite and FMC with addition of 10% SAP $% \left(\mathcal{S}_{1}^{2}\right) =0.00000000000000000000000000000000000$	22
WP4C Potential dew meter	24
Nilsson Soil Resistance Meter Model 400	25
Shrinkage plot for S300 with addition of 10% SAP	28
Shrinkage plot for Speswhite with addition of 10% SAP.	29
Shrinkage plot for FMC with addition of 10% SAP.	29
Soil water characteristic curve for S300	34
Soil water characteristic curve for Speswhite	35
Soil water characteristic curve for FMC	35
	Diagrammatic sketch of the structure of kaolinite Marconite installation in grounding system Schematic diagram of a dew-point Water PotentiaMeter, WP4 Superabsorbent polymer before oven drying Superabsorbent polymer after oven drying at 60° c Shrinkage limit test for S300, FMC and Speswhite with addition of 10% SAP Swelling index for S300, Speswhite and FMC with addition of 10% SAP WP4C Potential dew meter Nilsson Soil Resistance Meter Model 400 Shrinkage plot for S300 with addition of 10% SAP Shrinkage plot for Speswhite with addition of 10% SAP. Shrinkage plot for Speswhite with addition of 10% SAP. Shrinkage plot for FMC with addition of 10% SAP. Soil water characteristic curve for S300 Soil water characteristic curve for Speswhite Soil water characteristic curve for FMC

LIST OF SYMBOLS

Ω.m Ohm meter

LIST OF ABBREVIATIONS

ASTM	American Society for Testing and Materials
BS	British Standard
BET	Brunauer-Emmett-Teller
CEC	Cation Exchange Capacity
IEEE	Institute of Electrical and Electronics Engineers
SAP	Superabsorbent polymer
SWCC	Soil-Water Characteristic Curves

CHAPTER 1

INTRODUCTION

1.1 Background

Earth grounding system serves as lightning-protection in industrial and power plants (Liu et al., 2001). When grounding system is under fault, the system will separate the fault current to the earth by providing low resistivity path to decrease the earth potential rise at the local grounding system (Desmedt et al., 2001). Electrical conduction in rocks and soil can be divided into electronic conduction and ionic conduction where electronic conduction is the transfer of charges through a solid while ionic conduction is the transfer of ionic charges in a polar liquid such as rainwater (Laver et al., 2001). However, the current transmission is limited by the earth's resistance (PolyPhaser, 2010). Thus, it is vital to study the ground or the nature around the grounding system.

One of the major parameters to determine the performance of the grounding system is the resistivity of local soil (Dhrmadasa, 2011). Resistivity of soil varies with soil type, moisture content, temperature (Switzer, 1995), porosity, size of soil particles and so on (Fukue et al., 1999). Moisture content of soil helps the grounding electrode to disperse electrical current (Switzer, 1995). Site with low soil resistivity could install standard ground electrode system but site with high soil resistivity needs to increase the amount or diameter of grounding conductors or drive the ground electrodes deeper to achieve required ground electrode resistance. Normally, it is difficult to achieve the desired ground electrode resistance with the mentioned solutions. In these cases, ground enhancement materials or backfill material are used to enhance the grounding system to attain the required ground electrode resistance (Dale et al. 2017). A good ground enhancement material should provide low earth resistance over a long period with little variation of resistivity value (Gomes et al., 2010).

Bentonite is one of the suitable example of backfill material in decreasing and maintaining the low grounding resistance of electrodes (Lim et al., 2015) for a long time due to its high water absorption and retention tendency (Lim et al., 2013). It is a natural material that is composed predominantly of the clay mineral smectite which able to swell when in contact with free water (Keto, 2004). Bentonite is hydrate where it acts as drying agent to draw moisture from surrounding environment into its structure and holds the water chemically (Jones, 1980). According to research by Fukue et al. (1999), the resistivity for bentonite is high when water content is low. However, when the water content in bentonite is more than 40%, the resistivity is as low as 3Ω m. In Malaysia, it was reported that bentonite only deposited in several areas of Sabah such as in in Segama, Sepagaya, Mansuli and Andrassy (Samsuri, 2006). According to Malaysia which located one each in Perak, Johor and Selangor but the all the raw bentonite are imported. In 2010, Malaysia had imported 73,269 tonnes of bentonite mostly from India, China, USA, Australia and Germany which cost up to RM61,203,000.

Kaolinite is another type of clay which deposited in Malaysia at the state of Perak, Johor, Kelantan, Selangor, Pahang and Sarawak and around 112 million tons of kaolin have been discovered throughout the country (Baioumy et al., 2012). According to Malaysian Minerals Yearbook 2010, kaolin production in that year had increased to 530,331 tonnes from 487,632 tonnes recorded in the previous year mainly from Perak. However, kaolinite has lower plasticity behavior than bentonite. According to research by Horpibulsuk et al. (2011), bentonite has very high plasticity index which 175% is but the plasticity index for kaolin is very low which is 22%. Another study also shows that bentonite has 91.8% of plasticity index but kaolin only has 15.5% of plasticity index (Imai, 1980). Besides that, bentonite has higher optimum water content compare to kaolin due to bentonite's high sportive force from its surface electrical charges (Fattah et al., 2016). Fattah et al. (2016) state that bentonite has optimum water content up to 37% but kaolin only has 19.5% of optimum water content.

Superabsorbent polymer (SAP) is a hydrogel with three-dimensional polymer networks that expand when absorb water (Mudiyanselage et al., 2008). SAP is apply widely in several sector such as enhanced oil recovery, mine waste treatment, sludge dehydration, strengthening of concrete and soil amelioration (Gao, 2003). Gao states that SAP improve the soil quality for plant growth by absorbing water from rainfall or irrigation and releasing it slowly. SAP able to absorb more than hundred times of water than its own weight rapidly and retain water well even at high temperature and pressure (Guan et al., 2017).

In this study, 10% of SAP are mixed with the kaolinite clay to increase the water absorption and retention behavior. Plasticity and water suction behavior of enhanced kaolinite are studied and compared with kaolinite without and with 5% SAP and bentonite. Several tests such as liquid limit, plastic limit, shrinkage limit, specific gravity, swelling index, specific surface area, loss of ignition and cation exchange capacity are carried out to determine the improvement of plasticity of kaolinite at varying percentage of SAP and soil-water characteristics curves (SWCC) are established to study the water suction behavior of kaolinite. Lastly, resistivity of kaolinite at varying percentage of SAP is determined by four point probe method and compared with bentonite which normally use as earth grounding enhancement material.

1.2 Problem Statement

Bentonite is used as earth grounding backfill material to retain the moisture content of soil and to improve grounding effectiveness by lowering the soil resistivity. However, bentonite is only deposited in several areas at east of Sabah such as in Segama, Segapaya, Mansuli and Andrassy. All the bentonites used in Malaysia are imported from foreign country. A study on replacement of bentonite to kaolinite as earth grounding backfill material is carry out as the high availability of kaolinite in Malaysia such as in the states of Perak, Johor, Kelantan, Selangor, Pahang and Sarawak. But, kaolinite has lower plasticity and resistivity than bentonite. The most significant property of bentonite is its high water absorption and retention capacity. It can absorb up to hundred times of water of its own weight. Thus, superabsorbent polymer (SAP) may be added to the kaolinite to improve its plasticity and resistivity behavior by absorbing and retaining more water.

1.3 Research Objectives

The purpose of this study is to determine the effect of superabsorbent polymer (SAP) on the water absorption, plasticity and resistivity behavior of kaolinite:

1. To determine the improvement in the plasticity characteristic of kaolinite by addition of SAP at varying percentage.

2. To establish soil-water characteristics curve (SWCC) of kaolinite with addition of SAP at varying percentage.

3. To compare the resistivity of kaolinite at varying percentage of SAP.

1.4 Scope of Study

In this study, three kaolinite samples are used which are S300, Speswhite and FMC. The plasticity characteristic of these three kaolinites with 10% of superabsorbent polymer (SAP) is determined and compared with kaolinites with 5% SAP and without SAP, bentonite without SAP and macronite without SAP. Besides, soil-water characteristic curve (SWCC) of kaolinite with addition of SAP at varying percentage is established by using chilled mirror dew point method. Furthermore, a comparison on the resistivity of kaolinite at varying percentage of SAP is carried out by using four point probe method. Resistivity of bentonite is determined as a standard reference.

1.5 Significance of research

In this study, there will be some significant where addition of superabsorbent polymer (SAP) can improve the water absorption and retention, plasticity and resistivity characteristics of kaolinite. These improvement is important in replacement of bentonite which commonly used in geotechnical application with kaolinite.

1.6 Thesis overview

Chapter 2 present the overview of other research related to the study which include earth grounding system, clays, superabsorbent polymer (SAP) and soil water characteristic curve. Earth grounding system is explained on its importance and how the factors affecting soil resistivity influence the grounding resistivity. This chapter also overview about clays which function as backfill materials in earth grounding system. Clays that mention in this chapter include bentonite, kaolinite and macronite. Origin, structure, properties and applications of these clays are discussed. Furthermore, mechanisms of water absorption of SAP in modifying the plasticity and resistivity of kaolinite is explained. Formation, characteristics and uses of SAP also have been studied.

Lastly, soil water characteristic curve (SWCC) is presented in this chapter to illustrate the relation between soil water content and soil suction.

Chapter 3 present the experimental methodologies of this study which include the selection of materials such as S300, Speswhite and FMC, preparation of SAP and soil samples, physical properties, chemical property and resistivity of kaolinites when added 10% of SAP into the soil samples. Physical properties of modified kaolinite that to be determined include liquid limit (BS 1337: Part 2 1990: 4.3), plastic limit (BS 1377: Part 2: 1990: 5.3), shrinkage limit (ASTM D4943-08), specific gravity (BS 1377: Part2: 1990:8.3), swelling index (IS: 2720 (Part 40) 1977), specific surface area (ISO 9277:2010, DIN ISO 9227: 2013), loss of ignition (BS1377: Part 3: 1990: 4.3) and soil water characteristics curve (ASTM C387-99) while cation exchange capacity (ASTM D6836-07) is the chemical property to be determined. Lastly, resistivity test will be conducted by using four point probe method (ASTM G57) to determine and compare the resistance of bentonite and modified kaolinites which contain 5% and 10% of SAP.

Chapter 4 presents the results obtained from the laboratory tests. The results are compared and discussed by using 3 types of kaolinite added with 10% of SAP each. Besides, all the physical properties such as liquid limit, plastic limit, shrinkage limit, specific gravity, swelling index, specific surface area, loss of ignition and soil water characteristics curve, chemical property such as cation exchange capacity and resistivity are compared and explained with kaolinites without adding SAP, kaolinites added with 5% of SAP, bentonite and macronite. For example, the plastic limit for kaolinite with 10% of SAP is higher than kaolinite without SAP and 5% of SAP because addition of SAP improve the water absorption ability of kaolinites.

Chapter 5 present the conclusion of this study which include results and discussions, any important findings and recommendations should be made for next research.

CHAPTER 2

LITERATURE REVIEW

2.1 Introduction

This chapter will explain the earth grounding system in term of its importance and resistance. Besides, formation and application of bentonite and kaolinite clays are explained and their properties are compared. In this chapter, mechanisms, characteristic and usage of superabsorbent polymer (SAP) that affecting the plasticity of clay is also discussed. Lastly, soil water characteristic curve (SWCC) is explained to understand and compare the effect different content of SAP on three types of kaolinites.

2.2 Earth Grounding System

2.2.1 Importance of Earth Grounding System

The main purpose of earth grounding system is to minimize possible transient overvoltage (Adegboyega et al., 2011). When ground fault occur on transmission line, electrical or telecommunication towers around the fault are probable to gain high potentials to ground. Greatest danger to humans and animals may occurs when the tower reaches peak voltage (Endrenyil, 1967). Earth grounding system is needed so that the persons working in or around electrical or telecommunication towers are safe. Earth grounding system also able to decrease destruction to equipment in towers from lighting and thus decrease the disturbance to power system operations (Ahmeda, 2012). The grounding network resistance for power substation must be in 5 Ohm or less and for large station, the grounding resistance should be lower for satisfactory lightning protection (IEEE Std 142, 2007). Hence, inspection on tower potential conditions and protective effects of various transmission line grounding should be done timely (Endrenyil, 1967). Earth grounding system functions to separate fault currents aroused internally, externally

or by unbalance faults into the earth. Besides, all electric and electronic device can make up a large-scale system by referring to the extended grounding systems. (Menter et al., 1994). Ground electrode such as metal plate, metal pipe or metal conductor is driven into numerous places of earth to make electrical grounding connection. Normally, the ground electrode are made from copper, aluminum, mild steel or galvanized iron (Adegboyega et al., 2011).

2.2.2 Earth Grounding Resistance

Earth grounding resistance of the electrical or telecommunication tower should be effectively improved so that the lighting endure level and power supply reliability can be improved and at the same time the transmission line lighting trip-out rate and accident rate can be decreased (Feng et al., 2011). When lighting stroke to a tower, the high current will flows into ground through the tower footing and increase the soil ionization and thermal effects. This will decrease the ground resistance of the tower base (Harid et al., 2012). Normally, location with high transmission line tower grounding resistance has strong lighting activity. Terrain at this area is more complicated, soil resistivity is higher and the transport is not convenience (Feng et al., 2011). Earthing resistance of an electrode are affected by neighboring soil composition, soil temperature, soil moisture content, depth of electrode and severely soil resistivity (Adegboyega et al., 2011).

Soil resistivity is different from soil to soil due to different physical composition, moisture content, dissolved salt, particle size distribution and etc. (Parmar, 2011). Soil resistivity, ρ is measured in Ohm meters (Ω .m) corresponding to a 1 m² cylindrical cross sectional area and 1 m length. Soil resistivity shows how the soil conducts electric currents. Lower earth electrode resistance is needed for lower soil resistivity (Kižlo et al., 2009). All soils conduct electrical current but mostly have high resistivity or low electrical conductivity (Igboama et al., 2011). Table 2.1 shows the soil resistivity and earthing resistance required for different type of soil. Stoney soil has the highest soil resistivity which is 30000 Ω .m and Marconite has the lowest soil resistivity which is 0.001 Ω .m.

Type of soil	Soil resistivity (Ω.m)	
Farming soil loamy and clay soils	100	
Sandy clay soil	150	
Moist sandy soil	300	
Concrete 1:5	400	
Moist gravel	500	
Dry sandy soil	1000	
Dry gravel	1000	
Stoney soil	30 000	
Rock	107	
Bentonite	3	
Marconite	0.001	

Table 2.1Soil resistivity and earthing resistance required for different type of
soil

Source: Igboama et al. (2011)

Soil resistivity has most influence by moisture content of soil. The soil's moisture content is important because it helps chemicals in the soil that surround ground conductors carry the electrical current. (Slovick, 2017)Flow of electricity in the soil is greatly electrolytic as it is affected by the transport of ions dissolved in moisture (Kižlo et al., 2009). The soil resistivity should be low to avoid back flashover of lighting transmission line and to keep the ground potential rise in safety tolerance limits (Harid et al., 2012). Table 2.2 shows the effect of moisture content on soil resistivity. From the table, it is clearly shows that increasing of moisture content decrease the soil resistivity for top soil, sandy loam and silica based sand.

Moisture content by	Resistivity, Ω.m						
weight, %	Top soil	Sandy loam	Silica based sand				
0	1000000	1000000	-				
2.5	2500	150	3000000				
5	1650	430	50000				
10	530	185	2100				
15	210	105	630				
20	120	63	290				
30	100	42	-				

 Table 2.2
 Effect of moisture content on soil resistivity

Source: Harid et al. (2012)

Desired grounding resistance can be achieved by connecting a number of individual electrodes instead of using single low grounding resistance electrodes (IEEE Std 142, 2007). However, it is very difficult to get the expected grounding resistance by increasing the grid conductors as it may cost highly. In order to solve this problem, the soil surrounding electrodes should be modified by using conductive backfills. Bentonite

is one of the popular backfill material. By adding bentonite, grounding resistance may decrease up to 36% per rod (Kostic et al., 1999). According to IEEE Std 142-2007, ground resistance for large substations, transmission lines, or generating stations should not exceed 1 ohm while ground resistance for industrial plant substations and buildings and large commercial installations should be in 1 ohm to 5 ohm.

2.2.3 Grounding Enhancement Material

Nowadays, ground enhancing material or conductance-enhancement material is practice widely to decrease earth resistance due to high materials cost of earth conductors and limited space (Gomes et al., 2014). Ground enhancing material is usually termed as backfill materials. This material is laid inside the trench and the grounding electrode is installed. By this way, the soil resistivity surrounding the electrode is decreased to desired value and thus results in decreasing of grounding resistance value (Androvitsaneas et al., 2012). A good backfill material provide and maintain lower earth resistance than background soil for a long time and non-reactive with electrode (Gomes et al., 2014).

One of the most common backfill material is bentonite. Bentonite has well conductive property, high water absorption and retention as well as sticky property to most of the types of surface that it touches (Lim et al., 2013). However, bentonite must in wet condition to give low soil resistivity as bentonite move ions via ionic conduction in solution only. When bentonite loses moisture, its resistivity will increase (Switzer. 1995).

Marconite is a synthetic earth grounding material solely produced by James Durrans Group. It is an impure form of graphite that mix with cement to form an earth electrode backfill medium (Moore, 2000). According to James Durrans Group, Marconite is chemically inert and contains low sulphate and chloride, thus allow it to mix with all conventional types of cement. Marconite has 0.001 Ω m of resistivity and increse to 0.19 Ω m when mixed with cement.

Ground Enhancement Material (GEM) from ERICO is another alternative for backfill material(Switzer. 1995). Switzer states that GEM has more advantages over bentonite. GEM is maintanenece free as it is chemically stable and does not corrode ground electrode due to its low content of sulfate and chloride. According to ERICO, GEM powder provide resistivity less than 0.02 Ω m and this resistance is conctant throughout the life of grounding system once it is set.

Other material such as metal oxide, a waste product of steel industry is also proven as good backfill material. It possesses criteria of good backfill material (Lim et al., 2013) and less than 1% of corrosion on Galvanized Iron electrodes after more than two years in contact (Gomes et al., 2014). Natural material such as paddy dust or rice hulls and coco peat also can be used as earth grounding material. Janahitagama, a village in Western Province in Sri Lanka has apply rice hulls and coco peat as earth conductive material in three lightning protection towers each (Kumarasinghe, 2008). Kumarasinghe states that coco peat behave better as it can store higher moisture content. Another research by Jasni et al. (2010) shows that coconut coir peat and paddy dust do decrease the grounding resistance values but comparing to planting-clay soil and bentonite, their grounding resistance is higher. According to Eduful et al. (2009), palm kernel oil cake is another environmentally and economically friendly organic material that gives low resistance in effective resistance area. However, this material only suitable to the area relatively close to the earth electrode (Eduful et al., 2009)

2.3 Clay

Clays are fine-grained minerals which less than 2µm. Clays form naturally by crystallization in moisture condition on earth's surface and they are very abundant where they are main component of almost all sedimentary rocks which known as shales. Clays have unique properties such as cation exchange capabilities, plastic behavior when wet but harden when dry or fired, catalytic abilities, swelling behavior, and low permeability due to their distinct crystal structures (Guggenheim, 1995). These crystal structured are formed by planar arrangements of SiO4 structural units and many structural hydroxyls and water (Schulze, 2005).

2.3.1 Bentonite

Bentonite is a natural clay which form by years of volcanic action (Tu et al., 2006). It is formed by in situ devitrification of volcanic ash in water environment such as seawater, alkaline lakes or fresh water during or after deposition which contains highly colloidal and plastic montmorillonite clay mineral (Zuber et al., 2010). Bentonite is

noncorrosive, stable (Tu et. al., 2006), high cation exchange capacity, high bonding capacity, high plasticity and high propensity of reaction with organic compounds (Lim et al., 2013). When hydrated with water, bentonite is alkaline with 8 to 10 pH. Besides, bentonite has high water absorption capability as it is hydrophilic which strongly hydrated by water. Its volume will be enlarge to several times of original volume when water is absorbed by diffusion or capillary suction (Lim et al., 2013). According to Sridharan et al. (1999) and Mehyar et al. (2005), the water absorption of bentonite and kaolinite from different places is shown as Table 2.3.

-		-
Clay Type	Origin	Water Absorption (%)
Bentonite	Q'a Al Azraq, Jordan	207
Bentonite	Bulgaria	115
Bentonite	Indonesia	115
Bentonite	England	115
Fine Bentonite	USA	115
Bentonite	Kolar, India	292
Fine kaolinite	Bangalore, India	53.0
Fine kaolinite	Shimoga, India	47.1

Table 2.3Water absorption of bentonite and kaolinite from different places

Bentonite also has high capability to retain water for long period depending on temperature and atmospheric pressure (Lim et al., 2013). In addition, bentonite has thixotropy property where it perform like liquid when load apply by stirring or shaking and setting back to gel when standing (Teplitskiy et al., 2005). By adding bentonite, grounding resistance may decrease up to 36% per rod. Moreover, bentonite resistivity will become relatively low. At 300% moisture content, the resistivity is 2.5 Ω .m. Bentonite is suitable to be grounding backfill material especially in karst and sandy soils due to its good conductive properties, water absorption and retention abilities and surface adhesive (Kostic et. al., 1999).

Not only act as backfill material, bentonite also useful in supporting fluids. For example, bentonite is used sustain the sides of panel excavations for diaphragm walls in order to avoid runoff of fluid into the ground where the fluid pressure can withstand external pressure from soil and groundwater. Besides, bentonite also applies in other construction such as bored piles with big diameter, cut-off walls below ground and tunneling machines excavation. Its main function is to act as a barrier (Ball et al., 2006). Applications of bentonite not only limited in civil engineering but it is used widely in other field. Multitude uses of bentonite include drilling muds, foundry bonds, cat litter, ceramics, detergents, fertilizer, food additive, paint, paper, pharmaceuticals, sealants, water clarification and so on. (Murray, 2007)

2.3.2 Kaolinite

Name of kaolin was originated from Kauling or Gaoling, a village of Kiangsi province in China. Mining of this clay was first developed in the 11th century from the granitic hilly region near the Kauling village (Chen et al., 1997). This mining operated until 1964 before the clay is totally depleted (Haq et al., 2008). Kaolinite also deposited in Malaysia at the state of Perak, Johor, Kelantan, Selangor, Pahang and Sarawak. 12 million tons of kaolin have been discovered at these location. 341,223 tons of Kaolin is produced in 2007 and increased to 587,508 tons in 2006 (Baioumy et al., 2012).

Kaolinite is the most general kaolin mineral. Kaolins are rocks containing mainly one of the kaolin group minerals such as kaolinite, halloysite, dickite and nacrtie. Usually, kaolins are formed by alteration of Al silicate from feldspar mineral in comfortably high temperature and damp ambience (Murray, 1999). Primary kaolins are the aluminosilicate altered in situ by weathering, hydrothermal or sometimes by volcanic processes while secondary kaolins are sedimentary rock deposited in fresh or brackish water environment (Bloodworth et al., 1993).

Structure of kaolinite is the combination of one layer of octahedral sheet and one layer of octahedral sheet. This 1:1 layers are electrically neutral and stacked one above the other by hydrogen bonding (Schulze, 2005). According to Murray (2007), the structure of kaolinite is as shown in Figure 2.1.



Figure 2.1 Diagrammatic sketch of the structure of kaolinite

Normally, the color of kaolinite is white or almost while. Kaolinite has minimal layer charge and low base exchange capacity as it has few substitution in structural lattice. Comparing to bentonite, kaolinite lower surface area. With minimal layer charge and low surface area properties, kaolinite also has low absorption capacity (Murray, 2000). Kaolinite is relatively pure kaolin with 1.5 Mohs scale of hardness. Its softness and non-abrasiveness is suitable in industrial applications as it provide minimal wear and tear when in contact with equipment. By adding some chemical dispersant to negate the edge charges due to broken bond, kaolinite will disperse easily in water as it is hydrophilic. Thus, in water-based and paper-coating formulations, kaolinite will be added (Murray, 1997). Murray had summarize some comparison of kaolin (kaolinite) and smectite (bentonite) as shown in table 2.3.2.1.

Kaolin	Smectite
1:1 layer	2:1 layer
White or near white	Tan, olive, green, white
Little substitution	Octahedral and tetrahedral substitution
Minimal layer charge	High layer charge
Low base exchange capacity	High base exchange capacity
Pseudo- hexagonal flakes	Thin flakes and laths
Low surface area	Very high surface area
Very low absorption capacity	High absorption capacity
Low viscosity	Very high viscosity

Table 2.4Properties of kaolin (kaolinite) and smectite (bentonite)

2.3.3 Marconite

James Durrans and Sons Limited (2012) stated that Marconite is a synthetic earthing material that is specially manufactured for excellent electrical properties. Marconite is granular, dark grey in colour and dust free. It is a carbon by-product from oil refining. Marconite is expensive. Thus it only apply to small-scale applications such as electromagnetic shielding and anti-static flooring (Tuan et al., 2008). According to Thorne & Derrick UK, Marconite can conduct electricity without water. Unlike other traditional earthing materials such as Bentonite which conduct electricity through moving charged ions in the presence of water or salts, Marconite does not suffer effects from drying out. Marconite is a carbonaceous aggregate that replace normal aggregate in concrete mixing. It is more or less a conductive concrete with low resistivity Marconite has average 0.001 ohm-meter. Marconite cause little corrosion and able to perform well under hot weathers and rocky soils (Famous et al. 2015).

In the past year within UK, Marconite is used widely in latest power generation plants and rail engineers that does not require maintenance. Marconite is also use by worldwide. For example, Malaysia use Marconite in mass transit systems and Middle East apply Marconite in solar power farms (ETS, 2013). Figure 2.2 shows the Marconite installation in grounding system.



Figure 2.2 Marconite installation in grounding system

2.4 Superabsorbent Polymer (SAP)

Superabsorbent polymers (SAPs) or hydrogels are materials that can absorb and hold huge amounts of fluids without dissolving in water. Basically, they can absorb up to 15 times of fluids of their dried weight via osmotic pressure under load or without loud condition such as water, electrolyte solution, urine, brines, sweat, and blood. SAPs have loose cross-links and flexible polymer chains in 3-dimensional that carry dissociated and ionic functional groups (Kiatkamjornwong, 2007).

Mechanisms of water absorption of SAP can be categorized into physical and chemical absorption. Physical absorption can be further divided into several mechanisms. For example, for crystal structure such as silica gel and anhydrous inorganic salts, their structures are changed reversibly while for macro-porous structure such as soft polyurethane sponge, the water is absorbed via capillary force. There is also some combination of mechanisms such as absorption of water by capillary force is combined with hydration of functional groups (Zohuriaan-Mehr et al., 2008). Polymer backbone in SAP is hydrophilic which carboxylic acid groups (–COOH) that is water loving. Hydration or hydrogen bonds formation is occurred when water is added to SAP (Elliott, 2004). In chemical absorption mechanisms, the nature chemical absorbers such as metal

hydrides are converted via chemical reaction to absorb water (Zohuriaan-Mehr et al., 2008). Free energy between the chain networks of SAP and external solvent and the electrostatic repulsion give impetus to swelling behavior of SAP while elastic retractile response or elastic swelling of the SAP chain networks inhibit the SAP to swell. These three forces balance the water absorption in SAP (Mohan et al., 2006).

SAP is non-toxic, hydrophilic, biocompatible and has high swelling capacity in aqueous situation. Thus it is suitable for numerous field such as agriculture and horticulture, health, bioengineering, pharmaceutical, drug delivery, food industry, and other advanced technologies (Mohan et al., 2006). SAP is apply in construction to cure ultra-high performance concrete (Mechtcherine, 2012) and to self-heal concrete with crack width less than 50 to 100 micrometer (Shaikh, 2017). SAP also provide internal water resources to concrete as internal curing agent after the concrete settling. At the same time, SAP gives extra voids in the concrete mass to improve the concrete workability (Al-Nasra et al., 2013). According to Bhagat et al.(2016) SAP can be used in agriculture to reduce the water required by the crops and decrease the watering rate for crops at arid and desert areas. SAP is also ideal in water absorbing applications such as baby nappies and adults incontinence pads to absorb and retain large volume of aqueous solutions (Elliott, 2004). In oil and gas industry, SAP is used to avoid drilling fluids from diffusing into different layers in swellable packers and to enhance the oil extraction in oil recovery (Terhart, 2016).

2.5 Soil Water Characteristic Curve (SWCC)

Water content and its energy state characterized the water status in soil. The two principles form of energy are kinetic energy and potential energy (Or et. al., 2003). However, the kinetic energy is negligible as the water movement in soils is relatively slow. Difference in potential energy of water in soil tend to move from higher potential energy to lower potential energy until equilibrium (Hartmann, 1996). When water flow, larger pores in soil will drain first and the remaining pores will hold water more tenaciously. In this condition, the water in pores is under suction or negative hydrostatic pressure (Atkinson et. al., 2016).

Soil-water characteristic curve (SWCC) shows the relationship between water content and soil suction, where the water content is the water quantity contained in the pores of the soil (Fredlund, 1994) and the soil suction or partial vapor pressure of soil water is the free energy state of soil water (Padilla et.al., 2005). Padilla also states that water content in soil can be express in gravimetric water content, volumetric water content or degree of saturation while soil suction can be presented in matric suction or osmotic suction.

There are several methods to measure soil suction such as equilibration of small soil samples over salt solutions of known osmotic suction until water content are fixed, thermocouple psychrometry and measurement of water activity(Ebrahimi-Birang et.al., 2007). However, the first method may consume weeks to months to get final result while the second method is limited to suction of range -0.2MPa to -8.0 MPa (Gee et. al., 1992). Water activity meter is the device used to measure the water activity in the range of 0.1 to 1.0 (Gee et. al., 1992). Later, water activity meter is modified to dew-point Water PotentiaMeter (WP4), chilled-mirror dew-point psychrometer or chilled-mirror hygrometer to avoid the consequence of the temperature fluctuation of the surrounding on the suction (Ebrahimi-Birang et. al., 2007). Figure 2.3 shows a schematic diagram of a dew-point Water PotentiaMeter, WP4. The device consists of a mirror and photodetector cell, a temperature sensor, a fan and a sealed chamber.



Figure 2.3 Schematic diagram of a dew-point Water PotentiaMeter, WP4

Headspace of the sealed block chamber equilibrate the water potential of air in the chamber and the water potential or suction of the sample while chamber fan helps to accelerate the equilibration. The headspace vapour pressure is measured and the saturation vapour pressure is computed when water potential stable. In-built software will calculate the total suction of the soil specimen (in MPa and pF units) and show on the LCD panel of the WP4 along with the specimen temperature detected by the temperature sensor (Thakur et al., 2006).

SWCC is affected by initial water content and stress state but the effects decrease when suction increases (Zhou et. al., 2003). Other factor such as soil mineral composition, pore structure, soil body of contractility, and consolidation pressure also affected the characteristic curve directly or indirectly (Xu et al., 2013). According to research by Tripathy and others (2014), SWCC using chilled- mirror dew-point method is not affected by compaction factor. Compacted or uncompacted soil does not influence the relationship between water content and soil suction plot. Besides, dry unit weight of soil has not much influence on SWCC (Thakur et. al., 2006).

CHAPTER 3

METHODOLOGY

3.1 Introduction

In this study, several experiments are carried out to determine the physical properties, chemical property and resistivity of soil. Kaolinites added with 10% of superabsorbent polymer (SAP) are used as the soil samples. Physical properties of soil to be determined include liquid limit, plastic limit, shrinkage limit, specific gravity, swelling index, specific surface area and loss of ignition while cation-exchange capacity (CEC) is the chemical property of soil to be investigated. Besides, soil-water characteristic curve (SWCC) is also obtained to show the relation of water content and soil suction.

3.2 Selection of Material

3.2.1 Clay

Three types of kaolinites are tested throughout the study which include FMC, S300 and Speswhite. These kaolinite soils are readily available in Malaysia and thus the price is lower compare to bentonite soil.

3.2.2 Superabsorbent Polymer (SAP)

The superabsorbent polymer (SAP) is purchased from Daiso Japan.

3.3 Soil Sample Preparation

Superabsorbent polymer (SAP) is blended into small pieces and oven dry for 24 hours at 60 °C. Smash the SAP into powder form and add to the kaolinites. 10% of SAP is added and mix uniformly with kaolinites. These soil samples are kept in sealed plastic

bag and to be used for all the experiments. Figure 3.1(a) shows the SAP before oven drying and Figure 3.1(b) shows SAP after oven drying at 60° c.



Figure 3.1(a) Superabsorbent polymer before oven drying



Figure 3.1(b) Superabsorbent polymer after oven drying at 60° c

3.4 Physical Properties of Kaolinite

3.4.1 Liquid Limit, LL

Liquid limit of a soil is the empirically established moisture content in percentage of the weight of oven-dried soil at which change of consistency of soil from plastic to liquid state. Liquid limit test is carried out with cone penetration method according to BS 1337: Part 2 1990: 4.3.

3.4.2 Plastic Limit, PL

Liquid limit of a soil is the empirically established moisture content in percentage of the weight of oven-dried soil at which change of consistency of soil from semisolid to plastic state. It is the moisture content at which the soil just begin to crumble when rolling into a 3mm diameter thread on a ground glass plate. Plastic limit test is carried out according to BS 1377: Part 2: 1990: 5.3.

3.4.3 Shrinkage Limit

Shrinkage limit is the water content of soil when the soil is just saturated where the volume of water just exactly to fill all the pores of soil. Below shrinkage limit, reduction of water content will not affect the volume of soil. Swell- shrink test is conducted according to ASTM D4943-08. Figure 3.2 shows the soil samples S300, FMC and Speswhite with addition of 10% SAP after air dry. Shrinkage limit of soil is calculated with the following equation:

Shrinkage limit = W -
$$\left(\frac{V - V_s}{W_s}\right) \times 100\%$$
 3.1

Where W = Moisture Content of wet soil sample

V = Volume of wet soil sample

 $V_s = Volume of dry sample$

 W_s = Weight of oven dried soil sample



Figure 3.2 Shrinkage limit test for S300, FMC and Speswhite with addition of 10% SAP

3.4.4 Specific Gravity

Specific gravity or particle density of soil is the ratio of the weight of the given volume of soil sample to the weight of an equal volume of water. This test is carry out according to BS 1377: Part2: 1990:8.3. The specific gravity of soil is calculated using the following equation:

$$G_{s} = \frac{W_{2} - W_{1}}{(W_{4} - W_{1}) - (W_{3} - W_{2})}$$
3.2

Where $w_1 = mass of density bottle + stopper$

 $w_2 = mass of density bottle + stopper + dry soil$

 $w_3 = mass of density bottle + stopper + dry soil + distilled water$

 $w_4 = mass of density bottle + stopper + distilled water$

3.4.5 Swelling Index

Free swelling index is the increase in volume of soil when submerged in water without any restrictions. Free swelling index of soil is determined with free swelling method proposed by Holtz and Gibbs (1956).

Firstly, pour 10g of oven dried soil sample each into a 100ml measuring cylinder. The measuring cylinders are filled with distilled water until the 100ml mark of measuring cylinder. The soil-water mixture is left for swelling under controlled laboratory conditions so that the soil sample can gain equilibrium state of volume without any additional changes in the volume of soils. Figure 3.3 shows the swelling index test for S300, Speswhite and FMC with addition of 10% SAP. The final volume of the soils in each measuring cylinder is measured and the free swelling index of soil sample is calculated according the following equation:

Free Swell Index (%) =
$$\frac{V_1 - V_0}{V_0} \ge 100\%$$
 3.3

Where $V_1 =$ Soil volume after swelling, cm^3

 $V_0 =$ Volume of dry soil, $10cm^3$

Figure 3.3 Swelling index for S300, Speswhite and FMC with addition of 10% SAP3.4.6 Specific Surface Area

Specific surface area is the surface area of a soil over its mass or volume. Brunauer-Emmett-Teller (BET) method (ISO 9277:2010, DIN ISO 9227: 2013) is the most common and standardized method used to determine the specific surface area of soil.

Before the test started, the BET analyzer has to be warm up for at least 30 minutes. Firstly, fill the nanomaterials in the instrument specific glass holder and weight on a microbalance. Place 5g of soil sample in the instrument and heat up for at least 18 hours at 250 °C. Weight the soil sample again after the sample is cooled down to determine the possible mass losses. Next, the sample is place in the BET measurement unit for BET analysis. The soil sample will be cooled down to 77k which is the boiling point of liquid nitrogen. Lastly, nitrogen is injected under several pressures to determine the displacement of nitrogen gas for specific surface calculation.

3.4.7 Loss of Ignition

Loss of ignition test is carried out to identify the organic content in a soil sample by measuring the mass loss of soil when heated at high temperature. This test is done according to BS1377: Part 3: 1990: 4.3. The loss of ignition or organic content of soil is calculated as the following equation:

Loss of ignition (%) =
$$\left(\frac{m_1 - m_2}{m_1 - m_c}\right) \times 100\%$$
 3.4

Where $m_1 = Mass of crucible and oven-dry soil sample (g)$

 m_2 = Mass of crucible and soil sample after ignition (g)

 $m_c = Mass of crucible (g)$

3.5 Chemical Property

3.5.1 Cation-Exchange Capacity (CEC)

Cation exchange capacity (CEC) is the ability of soil to hold exchangeable positively charged ions. Soils contain higher clay fraction or organic matter tend to have higher CEC. CEC is tested by ammonium acetate method at pH7 according to ASTM C387-99.

3.6 Soil-Water Characteristic Curve (SWCC)

Soil-water characteristic curve (SWCC) shows the relationship between water content contained within the pores of soil and suction of soil. SWCC is plotted by chilled mirror dew point method according to ASTM D6836-07.

Firstly, 10 samples are prepared with 5g of soil mix with various water content from 0.1mL to 1.0mL. These samples are sealed in plastic bags to prevent evaporation of water and allow to equilibrate for 1 week. When the samples are ready to be tested, half fill the sample cups with samples. Make sure that the bottom surface of sample cups are fully covered by the samples. Figure 3.4 shows the device WP4C Dewpoint PotentiaMeter. WP4C Dewpoint PotentiaMeter is the device used to get the suction value of soil. The measurement also can be done when the device shows the $T_s - T_b$ value is 25.0. Next, place the sample cup in the drawer of WP4C and slide the drawer into the device. Turn the knob to left and wait until the data can be read and record the obtained suction value. Lastly, get water content of soil from its wet soil mass and dry soil mass. SWCC is plotted with soil suction value and soil water content.



Figure 3.4 WP4C Potential dew meter

3.7 Soil Resistivity

There are several methods to test for soil resistivity and one of the popular method is Wenner 4 Probe Test. This Wenner 4 Probe Test method (ASTM G57) was developed by Dr. Frank Wenner of the U.S. Bureau of Standards in year 1915 (Hurricane, 2013). Four electrodes are used in this method where two for current injection and two for voltage measurement. The device used to carry out the test is Nilsson Soil Resistance Meter Model 400. This device is shown in Figure 3.5. The soil resistivity is calculated as following equation:

$$\rho = 2\pi AR \tag{3.5}$$

Where ρ = average soil resistivity to a depth equal to A (ohm-cm)

A = probe spacing (cm)

R = soil resistivity measuring equipment's reading (Ohm)



Figure 3.5 Nilsson Soil Resistance Meter Model 400

CHAPTER 4

RESULTS AND DISCUSSION

4.1 Introduction

This chapter will discuss on all the results obtained from the laboratory test for 3 types of soil sample which are S300, Speswhite and FMC. These three samples are added with 10% of SAP and their properties are determined. These properties include Atterberg limit (liquid limit, plastic limit and shrinkage limit), specific gravity, swelling index, specific surface area, cation exchange capacity, loss of ignition and soil resistivity. Besides, soil water characteristic curve is plotted.

4.2 Physical properties

4.2.1 Liquid Limit

From cone penetration test, the liquid limit for S300, Speswhite and FMC with addition of 10% SAP are 460%, 495% and 500% respectively. From previous study, S300 has 41% of liquid limit and 83.85% with addition of 5% SAP. Meanwhile for Speswhite and FMC, the liquid limit with 5% SAP and without SAP are 51.4%, 96.11%, 80%, and 120% respectively. This clearly shows that addition of SAP has increase the liquid limit of kaolinite and the liquid limit increase with increasing proportion of SAP. For example, S300 with 10% SAP has increase its liquid limit by 1022% comparing to S300 without SAP. The liquid limit for S300 is lowest for 3 types SAP content follow by Speswhite and FMC. Higher liquid limit indicates stronger water absorption as surface charge between negative charge on clay surface and dipolar of water molecules increases (Sen, 2014). The liquid limit for bentonite is 437%. Addition of 10% SAP in kaolinite has achieved the desire liquid limit.

4.2.2 Plastic Limit

The plastic limit for S300, Speswhite and FMC with addition of 10% SAP are 150.38%, 141.45% and 275.30% respectively while the plastic limit are 48.79%, 37.8% and 64.87% respectively when added with 5% SAP. Plastic limit are the lowest for pure kaolinite where 27.9% for S300, follow by 32% for Speswhite and 49.4% for FMC. The plastic limit soil samples with 5% and 10% SAP increases according to the sequences of Speswhite, S300 and lastly FMC. Plastic limit for soil samples with 5% SAP is the lowest. Bentonite has 63% of plastic limit. S300 and Speswhite with 10% SAP and FMC with 5% and 10% SAP has higher plastic limit than bentonite. Plasticity of clay can be measured using Atterberg plasticity method in terms of its plasticity index, where plasticity index is the different between liquid limit and plastic limit. Increasing of plasticity index indicates increasing of plasticity of soil (Hall, 1923). The plasticity index for S300, Speswhite and FMC are 13.10%, 19.40% and 30.60% respectively while the plasticity index are 35.06%, 58.31% and 55.13% respectively when added with 5% SAP. The plasticity index of S300, Speswhite and FMC further increase to 309.62%, 353.55% and 224.70% respectively when added with 10% SAP. The plasticity index of bentonite is highest which is 374%.

4.2.3 Shrinkage Limit



Figure 4.1 shows the shrinkage plot for S300 with addition of 10% SAP.

Figure 4.1 Shrinkage plot for S300 with addition of 10% SAP

Void ratio increase with increase of wGs. The void ratio and shrinkage limit for S300 with 10% of SAP are 615.38 and 280%. Comparing to S300 and S300 with 5% SAP, which shrinkage limit are 38.5% and 40.0%, shrinkage limit of S300 with 10% SAP has increase greatly.

Figure 4.2 shows the shrinkage plot for Speswhite with addition of 10% SAP. Void ratio increase with increase of wGs. The void ratio and shrinkage limit for Speswhite with 10% of SAP are 500 and 235% while shrinkage limit for Spswhite and Speswhite with 5% SAP are 36.5% and 38.0%.



Figure 4.2 Shrinkage plot for Speswhite with addition of 10% SAP.



Figure 4.3 shows the shrinkage plot for FMC with addition of 10% SAP.

Figure 4.3 Shrinkage plot for FMC with addition of 10% SAP.

Void ratio increase with increase of wGs. The void ratio of FMC with 10% SAP is 526.63 and the shrinkage is limit for FMC with 10% of SAP is 241%, which is greater than shrinkage limit of FMC and FMC. Shrinkage limit for FMC 64.71%.

4.2.4 Specific Gravity

The specific gravity of kaolinites decrease when the proportion of SAP in kaolinite increase. When 5% of SAP is added to the kaolinite, the specific gravities are 2.70, 2.54 and 2.33 respectively for S300, Speswhite and FMC but the specific gravity decrease to 2.20, 2.13 and 2.17 respectively when the SAP added increase to 10%. The specific gravity for both pure S300 and FMC are 2.62 while specific gravity for pure Speswhite is 2.61. Bentonite shows the high specific gravity which is 2.80 while Marconite has lowest specific gravity which is 1.69. Specific gravity is closely related to mineralogy or chemical composition (Oyediran et. al., 2011), weathering history, where specific gravity increase with increasing of degree of weathering (Tuncer et.al, 1977) and appropriateness of soil as construction material, where soil with higher specific gravity gives higher strength for roads and foundations (Prakash et.al, 2002).

4.2.5 Swelling Index

S300, Speswhite and FMC have 1%, 0% and 10% of swelling index respectively and increase to 70%, 100% and 80% respectively with addition of 5% SAP. The swelling index increase greatly after adding 10% of SAP. The S300 with 10% SAP has 500% of swelling index while the swelling index of Speswhite with 10% SAP and FMC with 10% SAP are 440% and 470%. According to IS 1948, S300 and Speswhite without SAP have swelling index below 10% show that both of them have low degree of expansion. S300 and Speswhite with 5% SAP and FMC without SAP and with 5% SAP have medium degree of expansion as they have swelling indexes of 50% to 100%. S300, Speswhite and FMC with 10% SAP have swelling indexes above 200%. They are soils with very high degree of expansion. The major disadvantage of this method is crude as measuring 10cm³ of soil is not easy depending on personal judgement. It is normal to quantify 10cm³ of soil volume is equal to 10g of soil but this does not include consideration of density variation (Asuri et al., 2016). Holtz and Gibbs, 1956 stated that more than 100% of soil free swelling will cause lightly loaded structure damage but less than 50% of soil free swelling will not show significant volume changes. However, Dawson, 1953 reported that some Texas clays with free swelling values around 50% showed wide expansion as a result of extreme climate and expansion characters of the soil (Abdel, 2008).

4.2.6 BET Surface Area

BET surface area of S300 and Speswhite increase with addition of 5% SAP and 10%SAP. However, the BET surface area of FMC with 10% SAP decrease compare to FMC with 5% SAP. When S300 added with 5% SAP, the BET surface area is 1.4529 m²/g and increase by 27.59% when added with 10% SAP. Meanwhile, pure Speswhite has 7.8 m²/g of BET surface area and increase to 9.6513 m²/g and 11.2916 m²/g when added with 5% and 10% SAP. However, FMC with 5% SAP shows 9.6431 m²/g of BET surface area but the BET surface area decrease by 8.56% for FMC with 10% SAP. On the other hand, Bentonite shows the greatest BET surface area comparing to all kaolinite with and without SAP. Bentonite has 676 m²/g of BET surface area. Usually, nitrogen gas is for Brunauer-Emmett-Teller (BET) surface area analysis (Pennell, 2016). However, nitrogen does not penetrate the interlayer surfaces of soils and thus only external surfaces of soils are measured (Carter et al., 1986). Several assumption is made in determining the free surface area by BET method. For example, BET test only involve physical adsorption, the surface area of soil remain the same during adsorption process and the adsorbate in the soil does not form solution (Chiou, 1990)

4.2.7 Loss on Ignition (LOI)

LOI is the most common method to measure the soil organic matter content. The results show significant increases of loss on ignition with addition of SAP. Before applying SAP, the LOI of S300 is 0.0007%. After added with 5% of SAP, the LOI increase to 0.00785% and the LOI further increase to 8.722% when added with 10% SAP. While for Speswhite, the LOI before adding SAP and after adding 5% and 10% SAP are 0.086%, 0.022% and 9.128% respectively. On the other hand, LOI for pure FMC is 0.112% and the LOI increase to 0.056% and 9.572% when put in 5% and 10% SAP. Increase of soil organic matter indicates increase of water holding capacity of soil (Murphy, 2015). Soil organic matter also affect the cation exchange capacity of soil depending on the soil texture and pH range (Krull et. al, 2009). The results is affected by ignition temperature and heating time (Brix, 2008). Too low ignition temperature and too short burning time may not convert the organic matter completely into CO₂ but too high ignition temperature and too long heating time may convert, inorganic compounds such as carbonates and sulphate into CO₂ and SO₂.

4.2.8 Particle size distribution

Bentonite has 96.4 % of particle size that less than 2 μ m. This shows that Bentonite is a clay. Figure below shows the particle size distribution graph of Marconite. It is uniformly graded soil as most of the soil are the same size which is greater than 0.6mm but smaller than 3.35mm. D₁₀, D₃₀ and D₆₀ Marconite are 0.7mm, 0.93mm and 1.60mm respectively while the uniformity coefficient, Cu is 2.29 and the coefficient of gradation, Cc is 0.77.

4.3 Chemical Property

4.3.1 Cation-Exchange Capacity (CEC)

Cation exchange capacity (CEC) is a soil particle's ability to react with positively charged molecules such as nutrients, water, herbicides and other soil amendments (Goldy, 2011). CEC occurs when bonds around the crystal edges broken, substitute within lattice, and the hydrogen of exposed surface hydroxyls that may be replaced (Ma et al., 1999). The most common exchangeable cations in clay materials are Calcium (Ca⁺⁺), Magnesium (Mg⁺⁺), Hydrogen (H⁺), Potassium (K⁺), Ammonium (NH₄⁺) and Sodium (Na⁺) (Grim, 1953). Low CEC indicates that not much positively charged molecules are able to react with soil particle while high CEC shows that huge number of molecules can react with soil particle (Goldy, 2011). The CEC for S300, Speswhite and FMC with 10% of SAP are 460 meq/100g, 523.6 meq/100g and 344.1 meq/100g respectively. This shows great increase of CEC comparing to kaolinites with 5% SAP and without SAP. For example, S300 with 5% of SAP has 10.12 meq/100g of total CEC while S300 without SAP has only 1.80 meq/100g of total CEC. In all soil samples, the composition of Sodium ion is the highest. Bentonite has 90.31 meq/100g of total CEC.

Properties		S300			Speswhite			FMC		Bentonite	Marconite
-	Without SAP	With 5% SAP	With 10% SAP	Without SAP	With 5% SAP	With 10% SAP	Without SAP	With 5% SAP	With 10% SAP	Without SAP	Without SAP
Physical											
Properties											
Liquid limit (%)	41.00	83.85	460.00	51.40	96.11	495.00	80.00	120.00	500.00	437.00	-
Plastic limit (%)	27.90	48.79	150.38	32.00	37.80	141.45	49.40	64.87	275.30	63.00	-
Shrinkage limit	38.50	40.00	280.00	26.50	38.00	235.00	64.71	-	241.00	12.20	-
(%)											
Specific gravity	2.62	2.70	2.20	2.61	2.54	2.13	2.62	2.33	2.17	2.80	1.69
Swelling Index	1	70	500	0	100	440	10	80	470	-	-
(%)											
BET Surface Area	-	1.4529	1.8537	7.80	9.6513	11.2916	-	9.6431	8.8176	19.77	-
(m²/g)											
Loss on ignition	0.0007	0.00785	8.722	0.0860	0.02200	9.128	0.1120	0.0560	9.572	-	-
(%)											
Initial water	-	-	1.13	-	-	0.83	1.011	1.678	2.68	-	0.05
content (%)											
Particle size	-	-	-	-	-	-	-	-	-	< 2 µm:	< 63 µm:
districution										96.4 %	1.36%
Chemical											
Property											
Cation exchange	Na: 0.15	Na: 9.73	Na:	Na: 1.06	Na:18.30	Na:	Na:0.44	Na:14.73	Na:	Na:51.24	-
capacity	K: 0.12	K: 0.14	456.4	K: 0.37	K: 2.68	506.8	K: 0.21	K: 0.59	342.8	K: 1.40	
(meq/100g)	Mg: 0.05	Mg: 0.04	K: 1.6	Mg: 0.66	Mg: 0.07	K: 2.0	Mg:0.06	Mg: 0.62	K: 1.2	Mg: 9.43	
	Ca:0.482	Ca:0.192	Mg: 2.0	Ca: 1.09	Ca:0.867	Mg: 14.8	Ca:	Ca:	Mg: <0.1	Ca: 28.24	
			Ca: <0.5			Ca: <0.5	0.734	2.256	Ca: <0.5		
Total CEC (meq/100g)	0.81	10.10	460.0	3.19	21.92	523.6	1.45	20.90	344.1	90.31	

Table 4.1Properties of S300, Speswhite, FMC, Bentonite and Marconite with varying percentage of SAP

4.4 Soil Water Characteristic Curve

Figures 4.4 shows the soil water characteristic curve for soil sample S300 without and with 5% and 10% SAP.



Figure 4.4 Soil water characteristic curve for S300

S300 with 0% and 5% SAP do not show change in water content with increasing of soil suction but water content for S300 with 10% SAP decrease dramatically with increase of soil suction. The soil suction of S300 with 10% SAP varied between 3.93MPa to 109.99MPa corresponding to water content changes from 21.46% to 1.21%.

Figures 4.5 shows the soil water characteristic curve for soil sample Speswhite without and with 5% and 10% SAP. Based on the analysis, the water content for all soil samples decrease with increase of soil suction. The water content of 3 Speswhite samples decrease with increase of soil suction. At soil suction below 5.64MPa, water content of Speswhite with 10% SAP is higher Speswhite with 5% SAP but the water content of Speswhite with 5% SAP is higher Speswhite with 10% SAP above 5.64MPa soil suction. Water content for pure Speswhite is the lowest.



Figure 4.5 Soil water characteristic curve for Speswhite

Figures 4.6 shows the soil water characteristic curve for soil sample FMC without and with 5% and 10% SAP.



Figure 4.6 Soil water characteristic curve for FMC

Based on the analysis, the water content for all soil samples decrease with increase of soil suction. The water content of 3 FMC samples decrease with increase of soil suction. At all soil suction, water content of FMC with 10% SAP is higher than FMC with 5% SAP and pure FMC except at soil suction above 27.8MPa the water content of FMC with 5% SAP is slightly higher FMC with 10% SAP.

4.5 Soil Resistivity

The soil resistivity of various soil samples are as shown in Table 4.2.

Soil	Soil Resistivity (Ω.m)
S300	140
S300 + 5% SAP	130
S300 + 10% SAP	27
Speswhite	20
Speswhite + 5% SAP	26
Speswhite + 10% SAP	17
FMC	110
FMC + 5% SAP	110
FMC + 10% SAP	21
Bentonite	6.2
Marconite	0.38

Table 4.2Soil resistivity of various soil samples

From the result obtained, addition 5% of SAP has slightly decrease the soil resistivity of S300 and further decrease when 10% of SAP is added. The soil resistivity of S300, S300 with 5% SAP and S300 with 10% SAP are 140 Ω .m, 130 Ω .m and 27 Ω .m respectively. Soil resistivity of Speswhite is 20 Ω .m and increase to 26 Ω .m when added with 5% SAP but decrease to 17 Ω .m when added with 10% SAP. FMC and FMC with 5% SAP have same soil resistivity which is 110 Ω .m. The soil resistivity drop to 21 Ω .m when added with 10% SAP. S300 has highest soil resistivity follow by FMC and Speswhite. Overall, addition of SAP has decrease the soil resistivity of kaolinite. However, comparing to bentonite with 6.2 Ω .m and Marconite with 0.38 Ω .m soil resistivity, the desired resistivity for kaolinites are not achieved. According to IEEE Std 142-2007, Marconite is suitable as backfill material for large substations, transmission lines, or generating stations which grounding resistance should not exceed 1 Ω .m while bentonite is suitable for industrial plant substations and buildings and large commercial installations which grounding resistance should be in 1 Ω .m to 5 Ω .m. All the kaolinite samples do not meet the requirement for grounding resistance.

CHAPTER 5

CONCLUSION

5.1 Conclusion

The objectives of study are achieved. Based on the study, it can be conclude that:

- i. The plasticity index for S300, Speswhite and FMC are 13.10%, 19.40% and 30.60% respectively while the plasticity index are 35.06%, 58.31% and 55.13% respectively when added with 5% SAP. The plasticity index of S300, Speswhite and FMC further increase to 309.62%, 353.55% and 224.70% respectively when added with 10% SAP. Addition of 5% and 10% of SAP in S300 increase the plasticity index by 167.63% 2263.51% while the plasticity index of Speswhite increase by 163.85% and 1499.77% with addition of 5% and 10% Of SAP. FMC with 5% and 10% SAP increase their plastic limit by 80.16% and 634.31%. However, all of the plasticity index of kaolinites with or without SAP have lower plastic index than bentonite. The plasticity index of bentonite is highest, which is 374%.
- ii. Soil-water characteristics curve (SWCC) of kaolinite with addition of SAP at varying percentage is established. The water content for all soil samples decrease with increase of soil suction except S300 with 0% and 5% SAP do not show change in water content with increasing of soil suction. S300 with 10% SAP shows obvious reduction in water content with increase of soil suction comparing to S300 with 0% and 5% SAP. For pure Speswhite, the water content is lowest for all soil suction. On the other hand, at soil suction below 5.64MPa, water content of Speswhite with 10% SAP is higher Speswhite with 5% SAP but the water content of Speswhite with 5% SAP is higher Speswhite with 10% SAP above 5.64Mpa soil suction. At all soil suction, water content of FMC with 10%

SAP is higher FMC with 5% SAP and pure FMC except at soil suction above 27.8MPa the water content of FMC with 5% SAP is slightly higher FMC with 10% SAP.

iii. Addition of SAP has decrease the soil resistivity of kaolinites. S300 has highest soil resistivity follow by FMC and Speswhite. Addition 5% of SAP has slightly decrease the soil resistivity of S300 and further decrease when 10% of SAP is added. The soil resistivity of S300, S300 with 5% SAP and S300 with 10% SAP are 140 Ω .m, 130 Ω .m and 27 Ω .m respectively. Soil resistivity of Speswhite is 20 Ω m and increase to 26 Ω m when added with 5% SAP but decrease to 17 Ω m when added with 10% SAP. FMC and FMC with 5% SAP have same soil resistivity which is 110 Ω .m. The soil resistivity drop to 21 Ω .m when added with 10% SAP. Overall, addition of SAP has decrease the soil resistivity of kaolinite. However, comparing to bentonite with 6.2 Ω .m and Marconite with 0.38 Ω .m soil resistivity, the desired resistivity for kaolinites are not achieved. According to IEEE Std 142-2007, Marconite is suitable as backfill material for large substations, transmission lines, or generating stations which grounding resistance should not exceed 1 Ω .m while bentonite is suitable for industrial plant substations and buildings and large commercial installations which grounding resistance should be in 1 Ω .m to 5 Ω .m. All the kaolinite samples do not meet the requirement for grounding resistance.

5.2 Recommendations

- 1. Increase the percentage of SAP to 15%, 20% and so on in order to further enhance the properties of kaolinites.
- Crushing the SAP into smaller and regular shape in order to increase the result accuracy.
- 3. Modification of SAP to improve its absorbency, gel strength, and absorption rate.

REFERENCES

Abdel Latif, M. (2008). Swelling performance of some expansive soil treatment techniques.

Adegboyega, G. A., & Odeyemi, K. O. (2011). Assessment of Soil Resistivity on Grounding of Electrical Systems: A Case Study of North-East Zone, Nigeria. Journal of Academic and Applied Studies, 1(3), 28–38.

Ahmeda, M. (2012). Earthing performance of transmission line towers. Retrieved from http://orca.cf.ac.uk/42730/

Al-nasra, M., & Daoud, M. (2014). The Use of Super Absorbent Polymer as a Sealing Agent in Plain Concrete, 3(3), 132–137.

Androvitsaneas, V. P., Gonos, I. F., & Stathopulos, I. A. (2012). Performance of ground enhancing compounds during the year. 2012 31st International Conference on Lightning Protection, ICLP 2012. https://doi.org/10.1109/ICLP.2012.6344356

Asuri, S., & Keshavamurthy, P. (2016). Expansive Soil Characterisation: an Appraisal. INAE Letters, (April). https://doi.org/10.1007/s41403-016-0001-9

Atkinson, R. G., Brummel, D. A., Burdon, J. N., Patterson, K. J., & Schaffer, R. J. (2016). Plants in action. Plants in Action, 2–6.

Baioumy, H., & Ibrahim, A. R. (2012). Mineralogical Variations among the Kaolin Deposits in Malaysia, (Geos), 1–2. https://doi.org/10.5176/2251-3361

Bhagat, M. S., Ghare, A. D., & Ralegaonkar, R. V. (2016). Application of Super Absorbent Polymer in Flood Management and Agricultural Water Management, 1(1).

Bloodworth, A. J., Highley, D. E., & Mitchell, C. J. (1993). Industrial Minerals Laboratory Manual:Kaolin BGS TechnicalReport WG/93/1, 2014.

Brix, H. (2008). Soil Moisture and Loss on Ignition, 4-6.

Carter, D. L., Mortland, M. M., & Kemper, W. D. (1986). Specific Surface. Methods of Soil Analysis: Part 1 - Physical and Mineralogical Methods, 9(9), 413–423. https://doi.org/10.2136/sssabookser5.1.2ed.c16

Chen, P.-Y., Lin, M.-L., & Zheng, Z. (1997). On the origin of the name kaolin and the kaolin deposits of the Kauling and Dazhou areas, Kiangsi, China. Applied Clay Science, 12(1–2), 1–25. https://doi.org/http://dx.doi.org/10.1016/S0169-1317(97)00007-0

Chiou, C. T., Lee, J. F., & Boyd, S. A. (1990). The Surface Area of Soil Organic Matter. Environ. Sci. Technology, 24(8), 1164–1166.

D. J. Ball, Hutchinson, M. T., Jefferis, S. A., Shotton, P. G., Stansfield, L., & Wills, A.J. (2006). Bentonite Support Fluids in civil Engineering, (January), 1–27.

Dale R. Boling, P., ERICO®, & Solon, O. (2017). REQUIREMENTS FOR EARTHINGENHANCEMENTCOMPOUNDS.Retrievedfromhttps://www.erico.com/catalog/literature/CASE-EARTHING.pdf

Desmedt, M., Hoeffelman, J., & D.Halkin. (2001). Protecting Against Indirect Contacts in Hv Systems. October, (482), 18–21.

Dhrmadasa, I. T. (2011). Determination of earth resistivity profile in mul ti layer soil, Chapter 5 Earth Electrode Resistance, 78–105.

Ebrahimi-Birang, N., Fredlund, D. G., & Samarasekera, L. (2007). Hysteresis of the soilwater characteristic curve in the high suction range. Conference Proceedings of the The Diamond Jubilee: 60th Canadian Geotechnical Conference and 8th Joint CGS/IAH-CNC Groundwater Conference, 1061–1068. Retrieved from http://www.soilvision.com/subdomains/unsaturatedsoil.com/Docs/Research Papers/2007/Conference Papers/Hysteresis of the soil-water characteristic curve in the high suction range.pdf

Eduful, G., & Cole, J. E. (2009). Palm Kernel Oil Cake as an alternative to earth resistance-reducing agent. 2009 IEEE/PES Power Systems Conference and Exposition, PSCE 2009, 1–4. https://doi.org/10.1109/PSCE.2009.4840017

Elliott, M. (2004). Superabsorbent polymers. https://doi.org/10.1016/0003-2697(89)90590-3

Endrenyi, J. (1967). Analysis of Transmission Tower Potentials During Ground Faults. IEEE Transactions on Power Apparatus and Systems, PAS-86(10), 1274–1283. https://doi.org/10.1109/TPAS.1967.291904

ERICO. (2014). GEM - Ground Enhancement Material, 1–3. Retrieved from http://www.erico.com/products/gem.asp

ETS Cable Components (2013). A History Of Marconite Conductive Concrete. Retrieved from <u>https://www.etscablecomponents.com/2013/07/history-marconite-conductive-</u> <u>concrete/</u>

Famous, E., & Mnse, E. A. (2015). SOIL CONDITIONING FOR EARTHING BY, 1-6.

Fattah, M. Y., & Al-Lami, A. H. S. (2016). Behavior and characteristics of compacted expansive unsaturated bentonite-sand mixture. Journal of Rock Mechanics and Geotechnical Engineering, 8(5), 629–639. https://doi.org/10.1016/j.jrmge.2016.02.005

Feng, Z., Lu, L., & Feng, J. (2011). Research on reducing grounding resistance of transmission line tower grounding grid. 2011 International Conference on Electrical and

Control Engineering, ICECE 2011 - Proceedings, 1216–1219. https://doi.org/10.1109/ICECENG.2011.6057902

Fredlund, D. G., & Xing, A. (1994). Equations for the soil-water characteristic curve. Canadian Geotechnical Journal, 31(4), 521–532. https://doi.org/10.1139/t94-061

Fukue, M., Minato, T., Horibe, H., & Taya, N. (1999). The micro-structures of clay given by resistivity measurements. Engineering Geology, 54(1–2), 43–53. https://doi.org/10.1016/S0013-7952(99)00060-5

Gao, D. (2003). Superabsorbent Polymer Composite (SAPC) Materials and their Industrial and High-Tech Applications. https://doi.org/urn:nbn:de:swb:105-1484776

Gee, G. W., Campbell, M. D., Campbell, G. S., & Campbell, J. H. (1992). Rapid Measurement of Low Soil, 56(August), 1068–1070.

Goldy, R. (2011). What is your soil cation exchange capacity? Retrieved from http://msue.anr.msu.edu/news/what_is_your_soil_cation_exchange_capacity

Gomes, C., Ab Kadir, M. Z. A., Kottachchi, C. L., & Lim, S. C. (2014). Industrial wastes and natural substances for improving electrical earthing systems. International Journal of Electrical Engineering, 21(2), 39–47. https://doi.org/10.6329/CIEE.2014.2.01

Gomes, C., Lalitha, C., & Priyadarshanee, C. (2010). Improvement of earthing systems with backfill materials. 2010 30th International Conference on Lightning Protection, ICLP 2010, 1–9. https://doi.org/10.1109/ICLP.2010.7845822

Grim, R. E. (1953). Clay Mineralogy. McGRAW-HILL, 76(4), 412. https://doi.org/10.1097/00010694-195310000-00009 Guan, H., Li, J., Zhang, B., & Yu, X. (2017). Synthesis, Properties, and Humidity Resistance Enhancement of Biodegradable Cellulose-Containing Superabsorbent Polymer, 2017.

Guggenheim, S. (1995). Introduction to the properties of clay minerals, 371–388. https://doi.org/10.1002/9783527615292.ch6

Hall, F. P. (1923). Methods of measuring the plasticity of clays. [United States] Bureau of Standards. Technologic Papers, (no 234), 1 l., 345–366 tab., diagrs.

Harid, N., Griffiths, H., Ullah, N., Ahmeda, M., & Haddad, A. (2012). Experimental investigation of impulse characteristics of transmission line tower footings. Journal of Lightning Research, 36–44. https://doi.org/10.2174/1652803401204010036

Hartmann, R. (1996). Lecture Notes: Soil Physics, 74. Retrieved from http://www.iup.uni-heidelberg.de/institut/studium/lehre/bodenphysik/Outline-II/lecture_notes/soil_physics/I.pdf

Horpibulsuk, S., Yangsukkaseam, N., Chinkulkijniwat, A., & Du, Y. J. (2011). Compressibility and permeability of Bangkok clay compared with kaolinite and bentonite. Applied Clay Science, 52(1–2), 150–159. https://doi.org/10.1016/j.clay.2011.02.014

Hurricane. (2013). Wenner Four Pin Method for Soil Resistivity Test, 1–4.

IEEE. (2007). IEEE Std. 142. Grounding of Industrial and Commercial Power Systems (Vol. 2007).

Igboama, W., & Ugwu, N. (2011). Fabrication of resistivity meter and its evaluation. American Journal of Scientific and Industrial Research, 2(5), 713–717. https://doi.org/10.5251/ajsir.2011.2.5.713.717 Imai, G. (1980). Settling behaviour of clay suspension. Soils and Foundations, 20(2), 61– 77. https://doi.org/10.1248/cpb.37.3229

IS 1498. BUREAU OF UREAU OF INDIAN STANDARDS. Indian Standard Classification and Identification of Soils for General Engineering Purposes. BIS, New Delhi, 1970, (Reaffirmed 1987).

Ismail, S., Kasim, Z. C., Harun, H. C., & Aziz, A. K. M. T. A. R. A. (2010). Malaysian Minerals Yearbook 2010, 116.

James Durrans Group. (2012). Marconite - Description. Retrieved from http://www.marconite.co.uk/description.html

James Durrans Group. (2012). Marconite-Benefits, 44(172023), 370000.

Jasni, J., Siow, L. K., Kadir, M. Z. A. A., & Ahmad, W. F. W. (2010). NATURAL MATERIALS AS GROUNDING FILLER FOR LIGHTNING PROTECTION SYSTEM, 2010.

Jones, W. R. (1980). Bentonite rods assure ground rod installation in problem soils. IEEE Transactions on Power Apparatus and Systems, 75(4), 1343–1346.

Keto, P. (2004). Natural Clays as Backfilling Materials in Different Backfilling Concepts. POSIVA Oy Working Report 79, (March).

Kiatkamjornwong, S. (2007). Superabsorbent Polymers and Superabsorbent Polymer Composites. ScienceAsia, 33(s1), 39. https://doi.org/10.2306/scienceasia1513-1874.2007.33(s1).039

Kižlo, M., & Dqehujv, U. (2009). The Causes of the Parameters Changes of Soil Resistivity, 43–46.

Kostic, M. B., Radakovic, Z. R., Radovanovic, N. S., & Tomasevic-Canovic, M. R. (1999). Improvement of electrical properties of grounding loops by using bentonite and waste drilling mud. IEE Proceedings - Generation, Transmission and Distribution, 146(1), 1. <u>https://doi.org/10.1049/ip-gtd:19990057</u>

Krull, E. S., Skjemstad, J. O., & Baldock, J. a. (2009). Functions of Soil Organic Matter and the Effect on Soil Properties. GRDC Final Report CSO00029, 128. https://doi.org/GRDC Project No CSO 00029

Kumarasinghe, N. (2008). A low cost lightning protection system and its effectiveness. 20th International Lightning Detection Conference, 6. Retrieved from http://www.vaisala.com/Vaisala Documents/Scientific papers/A_low_cost_lightning_protection_system_and_its_effectiveness.pdf

Laver, J., & Griffiths, H. (2001). The variability of soils in earthing measurements and earthing system performance. Rev. Energ. Ren.: Power Engineering, School of ..., 57–61. Retrieved from http://www.cder.dz/vlib/revue/nspeciauxpdf/upec_7.pdf

Lim, S. C., Gomes, C., & Ab Kadir, M. Z. A. (2013). Characterizing of bentonite with chemical, physical and electrical perspectives for improvement of electrical grounding systems. International Journal of Electrochemical Science, 8(9), 11429–11447.

Lim, S. C., Gomes, C., & Ab Kadir, M. Z. A. (2013). Electrical earthing in troubled environment. International Journal of Electrical Power and Energy Systems, 47(1), 117–128. https://doi.org/10.1016/j.ijepes.2012.10.058

Lim, S. C., Gomes, C., Kadir, M. Z. A. A., Nourirad, G., & Malek, Z. A. (2015). Behaviour of backfill materials for electrical grounding systems under high voltage conditions. Journal of Engineering Science and Technology, 10(6), 811–826.

Liu, Y., Zitnik, M., & Thottappillil, R. (2001). An Improved Transmission-Line Model of Grounding System, 43(3), 348–355.

Ma, C. H. I., & Eggleton, R. a. (1999). Cation Exchange Capacity of Kaolinite. Clays and Clay Minerals, 47(2), 174–180. https://doi.org/10.1346/CCMN.1999.0470207

Mechtcherine, V., & Reinhardt, H. (2012). STAR 225-SAP Application of Superabsorbent Polymers (SAP) in Concrete Construction.

Mehyar, N., Madanat, M., & Sahawneh, J. (2015). Mineral status and future opportunity (Basalt). Retrieved from http://www.memr.gov.jo/EchoBusV3.0/SystemAssets/PDFs/AR/MineralTR/Basalt.pdf

Menter, F. E., & Grcev, L. (1994). EMTP-Based Model for Grounding System Analysis. IEEE Transactions on Power Delivery, 9(4), 1838–1849. https://doi.org/10.1109/61.329517

Mohan, Y., Keshava Murthy, P. S., Mohana Raju, K., & Sreedhar, B. (2006). Synthesis and Swelling Behavior of Acrylamide-Potassium Methacrylate Superabsorbent Copolymers. International Journal of Polymeric Materials, 55(1), 1–23. https://doi.org/10.1080/009140390901725

Moore, R. J. (2000). CAHORABASSA - APOLLO HVDC LINK R.

Mudiyanselage, T. K., & Neckers, D. C. (2008). Highly Absorbing Superabsorbent Polymer, (October 2007), 1357–1364. https://doi.org/10.1002/pola

Murphy, B. (2015). Key soil functional properties affected by soil organic matter - evidence from published literature. IOP Conference Series: Earth and Environmental Science, 25, 12008. https://doi.org/10.1088/1755-1315/25/1/012008

Murray, H. (1999). Applied clay mineralogy today and tomorrow. Clay Minerals, 34, 39–49. https://doi.org/10.1180/000985599546055

Murray, H. H. (2000). Traditional and new applications for kaolin, smectite, and palygorskite: A general overview. Applied Clay Science, 17(5–6), 207–221. https://doi.org/10.1016/S0169-1317(00)00016-8

Murray, H. H. (2007). Applied Clay Mineralogy - Occurrences, Processing and Application of Kaolins, Bentonites, Palygorskite-Sepiolite, and Common Clays. Developments in Clay Science (Vol. 2). <u>https://doi.org/10.1016/S1572-4352(06)02001-0</u>

Or, D., Tuller, M., Sciences, E., & Wraith, J. M. (2003). Soil water potential, (1985), 1– 13.

Oyediran, I. A., & Durojaiye, H. F. (2011). Variability in the Geotechnical properties of some residual clay soils from southwestern Nigeria, 2(9), 2–7.

Padilla, J. M., Perera, Y. Y., Houston, W. N., & Fredlund, D. G. (2005). A new soil-water characteristic curve device. Proceedings of Advanced Experimental Unsaturated Soil Mechanics - An International Symposium. Retrieved from http://books.google.com/books?hl=en&lr=&id=I2EX1H6nHFQC&oi=fnd&pg=PA15& dq=A+new+soil+water+characteristic+curve+device&ots=CY1uDhj1WU&sig=rFEDdr iUTCatwMAqx4-bASkyNkU

Parmar, J. (2011). Earthing in electrical network - purpose, methods and measurement, 1–11. Retrieved from http://electrical-engineering-portal.com/earthing-in-electrical-network-purpose-methods-and-measurement

Pennell, K. D. (2016). Specific Surface Area Area. Reference Module in Earth Systems and Environmental Sciences. Elsevier Inc. https://doi.org/10.1016/B978-0-12-409548-9.09583-X

PolyPhaser (2010). Grounding & Materials. Retrieved from http://www.polyphaser.com/SiteMedia/SiteResources/WhitePapersandTechnicalNotes/ 1485-035.pdf?ext=.pdf

Prakash S, Jain PK (2002). Engineering Soil Testing, Nem Chand & Bros, Roorkee, 26.

Samsuri, A. (2006). Mineralogyc And Physico-Chemical Characteristics Of Bentonite Clay From Sabah Malaysia. Environmental Science An Indian Journal, 1(1), 452. Retrieved from http://www.amazon.com/Environmental-Science-G-Tyler-Miller/dp/0495560162

Schulze, D. G. (200d). Clay minerals. Clay Minerals, 30, 246–254. https://doi.org/10.1081/E-ESS-120001688

Sen, B., & Pal, S. K. (2014). Index properties of soils collected from different locations and correlations of parameters. Electronic Journal of Geotechnical Engineering, 19 O, 3443–3452.

Shaikh, A., & John, R. (2017). SELF HEALING CONCRETE BY BACTERIAL AND CHEMICAL ADMIXTURES. International Journal of Scientific & Engineering Research, 8(3), 145–152.

Slovick, M. (2017). Conductive Concrete : A Key Element in any Lightning Protection Solution.

Sridharan, A., & Nagaraj, H. B. (1999). Absorption Water Content and Liquid Limit of Soils. Geotechnical Testing Journal, 22(2), 127–133. https://doi.org/10.1520/GTJ11271J

Switzer, K. (1995). Resistance. The Military Engineer, Volume 87(No.571).

Teplitskiy, A., Gee, R., & Kourmaev, R. (2005). Application of Physical-Chemical Properties of Bentonite Utilized In Construction, as viewed Through the TRIZ Prism.

The TRIZ Journal, (Oct), 1–6. Retrieved from http://www.triz-journal.com/application-physical-chemical-properties-bentonite-utilized-construction-viewed-triz-prism/

Terhart, M. (2016). Superabsorbent Polymers for Oil & Gas applications.

Thakur, V. K. S., Sreedeep, S., & Singh, D. N. (2006). Laboratory investigations on extremely high suction measurements for fine-grained soils. Geotechnical and Geological Engineering, 24(3), 565–578. https://doi.org/10.1007/s10706-005-1147-

Thorne & Derrick UK. (n.d.). Marconite. Retrieved from http://www.cablejoints.co.uk/upload/Marconite-Aggregate---Brochure.pdf

Tripathy, S., Elgabu, H., & Thomas, H. (2014). Soil-water characteristic curves from various laboratory techniques. Unsaturated Soils: Research & Applications, (September), 1701–1707. <u>https://doi.org/10.1201/b17034-248</u>

Tu, Y., He, J., & Zeng, R. (2006). Lightning impulse performances of grounding devices covered with low-resistivity materials. IEEE Transactions on Power Delivery, 21(3), 1706–1713. https://doi.org/10.1109/TPWRD.2006.874110

Tuan, C. Y. (2008). Implementation of Conductive Concrete For Deicing, 1(July).

TUNCER, E. R., & LOHNES, R. A. (1977). AN ENGINEERING CLASSIFICATION FOR CERTAIN BASALT-DERIVED LATERITIC SOILS E.R., 11, 319–339.

Xu, X., Zhang, F. Y., & Gu, M. (2013). Experimental Research on Soil-Water Characteristic Curve of Unsaturated Soils. Applied Mechanics and Materials, 353–356, 554–557. https://doi.org/10.4028/www.scientific.net/AMM.353-356.554

Zhou, J., & Yu, J. (2005). Influences affecting the soil-water characteristic curve. Journal of Zhejiang University SCIENCE, 6A(8), 797–804. https://doi.org/10.1631/jzus.2005.A0797 Zohuriaan-Mehr, M. J., & Iran, K. K. (2008). Superabsorbent Polymer Materials: A Review. Iranian Polymer Journal, 17(6), 451–477. https://doi.org/http://journal.ippi.ac.ir

Zuber, M., Zia, K. M., Mahboob, S., Hassan, M., & Bhatti, I. A. (2010). Synthesis of chitin-bentonite clay based polyurethane bio-nanocomposites. International Journal of Biological Macromolecules, 47(2), 196–200. https://doi.org/10.1016/j.ijbiomac.2010.04.022